Author's response to reviews

Title: Comparative health systems research in a context of HIV/AIDS: Lessons from a multi-country study in South Africa, Tanzania and Zambia

Authors:

Suraya S Dawad (dawads1@ukzn.ac.za)
Nina N Veenstra (veenstran@ukzn.ac.za)

Version: 3 Date: 22 October 2007

Author's response to reviews:

Formatting changes have been made as follows:

1. The text 'Title' was removed from above the title.
2. Section numbers were removed.
3. The title for the introduction and references sections were duplicated by mistake and so this was rectified.
4. The table was moved to a position after the references.
5. The section on abbreviations was removed, since there are no abbreviations in the text.